Author’s response to reviews

Title: Ectopic Insulinoma: Case Report

Authors:

Mengqing Sun (sunmq1128@126.com)
Yaping Luo (luoyaping@live.com)
Yan You (youyan@pumch.cn)
Xianlin Han (hanxianlin@pumch.cn)
Yupei Zhao (zhao8028@263.com)

Version: 3 Date: 28 Nov 2019

Author’s response to reviews:

Dear Editor,

The statement "Written informed consent for publication of their clinical details and/or clinical images was obtained from the patient" has been included in the Declarations part.

You suggested us to provide clarification on YPL and YY’s contributions, or remove their names from the list of authors and place them in the “Acknowledgements” section instead. In this case report, YPL has made an important contribution to the patient's preoperative localization of the tumors, and helped to write an interpretation of the nuclear medicine image. YY has made an important contribution to the postoperative pathological diagnosis of patients, and helped to write the pathological diagnosis part. We think they deserve to be authors.

Thank you!